Author’s response to reviews

Title: Economic hardships in adulthood and mental health in Sweden. The Swedish National Public Health Survey 2009.

Authors:

Johanna Ahnquist (johanna.ahnquist@ki.se)
Sarah P Wamala (sarah.wamala@ki.se)

Version: 6 Date: 5 October 2011

Author’s response to reviews:

Dear BMC Series Editorial Production Team,
We have made a few, very small, typological changes to the paper.
Kind regards,
Johanna Ahnquist